Author’s response to reviews

Title: Exploring the relationships between International Classification of Functioning, Disability and Health (ICF) constructs of Impairment, Activity Limitation and Participation Restriction in people with osteoarthritis prior to joint replacement.

Authors:

Beth Pollard (beth.pollard@abdn.ac.uk)
Marie Johnston (m.johnston@abdn.ac.uk)
Paul Dieppe (paul.dieppe@pms.ac.uk)

Version: 10 Date: 2 May 2011

Author’s response to reviews:

Minor typographic errors have been corrected. The additional files still had tracked changes from the previous version so initially I accepted all changes and then made the additional changes.